Author's response to reviews

Title: Circulating Immune Complexes and trace elements (Copper, Iron and Selenium) as markers in oral precancer and cancer: A randomised controlled clinical trial

Authors:

Sunali S Khanna (sunali3011@yahoo.com)
Freny R Karjodkar (fkarjodkar@yahoo.co.in)

Version: 6 Date: 25 July 2006

Author's response to reviews:

Dear sir,

This is the version which has been altered in accordance with the corrections you said to make. In the introduction some lines have been deleted. Material and methods have only the required details. Discussion has been made more extensive and statistics and table parts have been condensed. Kindly accept this and tell me any other corrections.

Thank you,

author